Author's response to reviews

Title: Unusual cause of a painful right testicle in a 16-year-old male: a case report

Authors:

Khalid N Shehzad (khalidshehzad@hotmail.com)
Amjid A Riaz (mrariaz@hotmail.com)

Version: 3 Date: 14 October 2010

Author's response to reviews:

To

The Editor,

Journal of Medical Case Reports.

Dear Sir/Madam,

Thank you very much indeed for accepting our manuscript entitled 'Unusual cause of a painful right testicle in a 16-year-old male: a case report' for publication in principle.

As advised, we have made the following changes:

1) Authors’ names are now in the 'First name, initial, Surname' format.
2) Reference citations are now in the format that you suggested i.e. [1], [2] etc.
3) 'Boy' has been replaced with 'male' in both the abstract and main text.

We have both read through the revised manuscript and approve it.

Kind regards,

Khalid N Shehzad

Amjid A Riaz